Author’s response to reviews

Title: Nutrition in primary and secondary prevention of cardiovascular risk in the continental and Mediterranean regions of Croatia

Authors:
Jozica Sikic (josicas1@gmail.com)
Mira Stipcevic (mira.stipcevic@gmail.com)
Hrvoje Vrazic (vrazic@gmail.com)
Jasna Cerkez Habek (jasna.habek@gmail.com)
Eduard Margetic (eduard.margetic@gmail.com)
Dario Gulin (dariogulin@gmail.com)

Version: 3 Date: 31 Aug 2017

Author’s response to reviews:

Dear Editor-in-chief of BMC Cardiovascular Disorders,

Thank you for all comments that helped in improvement of this manuscript.

Here are point-by-point responses to the comments:

Editor comments-
- Limitations are added in Discussion section.

With regards

Corresponding author
Dario Gulin, MD